Craig Chambers CSE 501 Reading Assignments

Reading Assignment, Week #3

For: Read: 1/18 holiday 1/20 [Tjiang & Hennessy 92], [Chambers et al. 96] 1/22 Sections 19.Intro, 19.1, 19.3, 19.4, 19.6 of text; skim [Alpern et al. 88]

Bibliography:

[Tjiang & Hennessy 92] Steven W.K. Tjiang and John L. Hennessy. Sharlit - A Tool for Building Optimizers. In PLDI ’92.

[Chambers et al. 96] Craig Chambers, Jeffrey Dean, and David Grove. Frameworks for Intra- and Interprocedural Dataflow Analysis. UW TR.

[Alpern et al. 88] Bowen Alpern, Mark N. Wegman, and F. Kenneth Zadeck. Detecting Equality of Variables in Programs. In POPL ’88.